Author's response to reviews

Title: Protective effects of Egyptian cloudy apple juice and apple peel extract on lipid peroxidation, antioxidant enzymes and inflammatory status in diabetic rat pancreas

Authors:

Samah M Fathy (samahfathy1972@gmail.com)
Ehab A Drees (ehab.kadry@gmail.com)

Version: 5 Date: 3 November 2015

Author's response to reviews: see over
Dear Editor

Thank you for the time you spent to handle our manuscript.

I declare that I uploaded the revised manuscript for the article “Protective effects of Egyptian cloudy apple juice and apple peel extract on lipid peroxidation, antioxidant enzymes and inflammatory status in diabetic rat pancreas”.

First, for the minor revisions by referee 2:

Target peaks were still hard to recognized in fig 1 and 2, I am wondering whether the author can provide extracted ion chromatogram for each target peaks.

**The answer:**

That is why we split fig 1 (Gas Chromatography mass spectral analysis of Egyptian CAJ showing the mass spectral analysis of polyphenolic compounds) into fig 1 and 2 in the revised manuscript. In addition, it is further confirmed by table 1. By the same way, fig 2 (Gas Chromatography mass spectral analysis of Egyptian APE showing the mass spectral analysis of polyphenolic compounds) split into fig 3 and 4 in the revised manuscript. Also, it is further confirmed by table 2.

Second, for the journal style:
1. We added Competing interests section.
2. References were adjusted according to the journal format.

I hope my answers will be appropriate and satisfactory.

Thanks for the time you spend in reviewing my article and I am looking forward to hear from you.

Sincerely

Samah Fathy                                             Permanent Address
Biology Department                                     Zoology Department
Faculty of Science                                      Faculty of Science
Jazan University                                        Fayoum University
Jazan                                                   Fayoum
KSA                                                    Egypt